Author’s response to reviews

Title: Methionine synthase A2756G polymorphism may predict ulcerative colitis and methylenetetrahydrofolate reductase C677T pancolitis, in Central China

Authors:

Min Chen (chenmin8106@yahoo.com.cn)
Laurent Peyrin-Biroulet (peyrin-biroulet@netcourrier.com)
Bing Xia (bingxia2004@yahoo.com.cn)
Rosa-Maria Rodriguez-Gueant (rm.rodriguez@chu-nancy.fr)
Jean-Pierre Bronowicki (jp.bronowicki@chu-nancy.fr)
Marc-Andre Bigard (ma.bigard@chu-nancy.fr)
Jean-Louis Gueant (Jean-Louis.Gueant@medecine.uhp-nancy.fr)

Version: 5 Date: 20 July 2008

Author’s response to reviews:

Dear Editor,

please find enclosed our manuscript with the pending formatting requirements.

Thank you for accepting to publish our article in BMC med genet.

Sincerely yours,

JL Guéant, MD, PhD, AGAF